Author's response to reviews

Title: En bloc Extended Total Thymectomy and Extrapleural Pneumonectomy in Masaoka stage IVA Thymomas

Authors:

Jhingook Kim (jkimsmc@skku.edu)
Hee Chul Yang (hcyang@snubh.org)
Hong Kwan Kim (hkts@skku.edu)
Yoo Sang Yoon (v2sky@hotmail.com)
Yong Soo Choi (choi.smcts@gmail.com)
Kwhanmien Kim (kmkim0070@yahoo.co.kr)
Young Mog Shim (ymshim@skku.edu)
Jungho Han (hanjho@skku.edu)

Version: 2 Date: 4 March 2011

Author's response to reviews: see over
March 3, 2011

Vipin Zamvar, Editor
The Journal of Cardiothoracic Surgery
Royal Infirmary of Edinburgh, United Kingdom

Dear Mr. Zamvar:

Enclosed please find a revised manuscript entitled “En bloc Extended Total Thymectomy and Extrapleural Pneumonectomy in Masaoka stage IVA Thymomas” (Original manuscript: MS: 4448957449000991) by Kim et al, which we wish to submit to the Journal of Cardiothoracic Surgery for publication as an Original Research Article.

I appreciate very much that you have read my manuscript and recommended that our manuscript be published. And I apologize for being late in submitting the revised manuscript. My co-authors and I have read through the reviewer’s comments and we were very impressed by them. Thanks to the comments, we were able to improve our manuscript and we appreciate it very much.

All the authors of this manuscript have agreed for this revision. If you have any questions regarding this manuscript, please contact me.

If you could kindly review this revised manuscript in consideration of its publication in the JCTS, we are very grateful. I look forward to hearing from you.

Sincerely yours,

Jhingook Kim, M.D.
Professor